Author's response to reviews

Title: Decreased Hsp90 expression in infiltrative lobular carcinoma: an immunohistochemical study.

Authors:

Flora Zagouri (florazagouri@yahoo.co.uk)
Theodoros N. Sergentanis (tsergentanis@yahoo.gr)
Afrodite Nonni (afnonni@med.uoa.gr)
Christos A. Papadimitriou (chr_papadim@yahoo.gr)
Anastasia Pazaiti (breastumit_ipp@yahoo.gr)
Nikolaos V. Michalopoulos (nmichal@yahoo.gr)
Panagiotis Safioleas (panossafioleas@yahoo.gr)
Andreas Lazaris (alazaris@med.uoa.gr)
George Theodoropoulos (georgetheocrs@live.com)
Effstratios C. Patsouris (efpatsouris@med.uoa.gr)
George Zografos (gzografo@med.uoa.gr)

Version: 6 Date: 6 August 2010

Author's response to reviews: see over
George C. Zografos  
Professor of Surgery, University of Athens  
101, Vas Sofias Ave,  
Ampelokipi, Athens 11521, Greece.  
e-mail: gzografo@med.uoa.gr  
tel.number: +302106426390  
fax: + 302106426390  
Mobile: +306932524836  

To  
The Editorial Board of *BMC Cancer*  

Athens, 06/08/2010  

Dear Editor,  

I am sending for publication to your distinguished Journal the revised manuscript entitled: “Decreased Hsp90 expression in infiltrative lobular carcinoma: an immunohistochemical study.”  

We have made all the changes indicated, so that the manuscript conforms with the style of your journal. Please find a point-to-point description of the changes, in the next page.  

Yours sincerely,  

George C. Zografos,  
Professor of Surgery  
University of Athens
We would like to thank you for having our manuscript accepted. Following your comments a point-by-point description of the changes follows.

1. The author list in the final version of the manuscript has been written exactly as in the submission system.
2. In the Authors’ contribution section, a sentence has been added, declaring that “all authors have read and approved the final manuscript”.
3. In the affiliation section, the city has been included.
4. The key-words have been removed from the abstract.
5. The Authors’ contribution has been presented as an abstract.
6. Reference 19, has been properly cited.
7. The visible vertical lines have been removed from the Tables.
8. The table title has been appeared in one-two sentence long, above the Table.
9. The word “Figure” has been written in one way throughout the whole manuscript.
10. The manuscript has been checked for typographical errors.